Author’s response to reviews

Title: Lack of Association of the Opioid Receptor mu 1 (OPRM1) A118G Polymorphism (rs1799971) with Alcohol Dependence: Review and Meta-analysis of Retrospective Controlled Studies

Authors:

Xiangyi Kong (Xiangyi_Kong@hms.harvard.edu)
Hao Deng (XKONG@mgh.harvard.edu)
Shun Gong (gongshunmmmu@foxmail.com)
Theodore Alston (Alstoncorbett@comcast.net)
Yanguo Kong (kongyanguo1@gmail.com)
Jingping Wang (JWANG23@mgh.harvard.edu)

Version: 3 Date: 06 Oct 2017

Author’s response to reviews:

Dear the editor

Thank you for your comments. We have revised the article point by point. The main changes and the responses to the reviewers are as follows:

1. It has been noted that there is some overlap with one of your previous publications. While we appreciate that you may want to use some of these ideas, please reword some of the highlighted sections in the attached CrossRef report, as we cannot publish manuscripts with text that has been published previously. Please note that the software used to detect overlaps is imperfect and highlights two or more consecutive matching words, therefore, please just focus on those whole sentences/paragraphs highlighted.

Corrected as suggested.

2. Please change the Background and Object heading in the Abstract to Background.

Corrected as suggested.
3. Please change the Introduction heading to Background.
Corrected as suggested.

4. Please clarify in the Ethics approval and consent to participate section who waived the need for ethics approval and consent, was this waiver granted by an International Review Board, or is this due to a law or national guideline. If so, please state specifically which one and provide a link or citation.
Corrected as suggested.

5. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship
Corrected as suggested.

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Corrected as suggested.

6. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.
Corrected as suggested.

7. Please remove any supplementary files which you do not want to be published, such as the response letter, as they are no longer needed at this stage.
Corrected as suggested.
8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Corrected as suggested.

Many thanks!

Best wishes,

Yanguo Kong